Author's response to reviews

Title: Variability of the Impact of Adverse Events on Physicians' Decision Making

Authors:

Ralucu Cozmuta (ralluca.cozmuta@yale.edu)
Elizabeth R Wahl (erwahl@gmail.com)
Peter A Merkel (pmerkel@upenn.edu)
Liana Fraenkel (liana.fraenkel@yale.edu)

Version: 3 Date: 22 May 2014

Author's response to reviews:

Response to the Reviewer's Comments

I have removed the abbreviations from the abstract and also reduced the abbreviations used in the main text.
The Background section is now clarified in the abstract.
The authors' contributions' are included at the end of the manuscript text before the references.
The ethics committee approval has been clarified.